Author's response to reviews

Title: Cluster randomised trial in the general practice research database: 1. Electronic decision support to reduce antibiotic prescribing in primary care (eCRT study)

Authors:

Martin C Gulliford (martin.gulliford@kcl.ac.uk)
Tjeerd van Staa (Tjeerd.VanStaa@mhra.gsi.gov.uk)
Lisa Mcdermott (l.mcdermott@soton.ac.uk)
Alex Dregan (alexandru.dregan@kcl.ac.uk)
Gerard McCann (Gerard.McCann@mhra.gsi.gov.uk)
Mark Ashworth (mark.ashworth@kcl.ac.uk)
Judith Charlton (judith.charlton@kcl.ac.uk)
Andrew P Grieve (grieve@clinresearch.com)
Paul Little (p.little@soton.ac.uk)
Michael V Moore (lmvm198@soton.ac.uk)
Lucy Yardley (l.yardley@soton.ac.uk)
electronic Cluster Randomised Trial Research Team eCRT Research Team (martin.gulliford@kcl.ac.uk)

Version: 5 Date: 10 May 2011

Author's response to reviews:

Authors’ contributions section has been revised.